Author’s response to reviews

Title: A Point-of-Care test for facing the burden of undiagnosed celiac disease in the Mediterranean area: a pragmatic design study

Authors:

Stefano Costa (stefan.costa@gmail.com)
Luca Astarita (luca_aSTARITA@hotmail.com)
Mongi Ben-Hariz (mongi.benhARIZ@RNS.tN)
Giovanni Curro’ (giovanni.curro75@gmail.com)
Jernej Dolinsek (jernej.dolinsek@UKC-MB.SI)
Aydan Kansu (aydANKANSU@gmail.com)
Giuseppe Magazzu’ (magazzug@unime.it)
Stefania Marvaso (stefaniamarvaso@tiscALi.it)
Dusanka Micetic-Turku (dusanka.turk13@gmail.com)
Salvatore Pellegrino (salvatOREPellegrino2014@gmail.com)
Giuseppe Primavera (beppeprimavera@virgilio.it)
Pasqualino Rossi (P.rossi@sanita.it)
Andrea Smarrazzo (and1_mlK@hotmail.it)
Francesca Tucci (ftucci84@gmail.com)
Carmela Arcidiaco (margherita@unina.it)
Luigi Greco (ydongre@unina.it)

Version: 2
Date: 15 September 2014

Author’s response to reviews:

The affiliation of Pasqualino Rossi was changed

The sentence “Written informed consent for participation in the study was obtained from participants or, where participants were children, a parent or guardian” was added in the Methods section

Disclosure of funding sentence was changed